Title: Adherence to antiretroviral therapy and associated factors among HIV infected children in Ethiopia: unannounced home-based pill count versus caregivers’ report

Authors:

Silenat Birosaw (silenatbw@yahoo.com)
Woldaregay Erku Abegaz (woldearegay.erku@aau.edu.et)
Markos Abebe (markosabebe@yahoo.com)
Workeabeba Abebe (workeabebasol@yahoo.com)
Mulugeta Belay (mulg2002@yahoo.com)

Version: 7 Date: 22 August 2013

Author’s response to reviews:

You provided a covering letter for the most recent version of this manuscript on 19 August 2013:

Authors’ response to Reviewers’ comments (Version III)

Adherence to antiretroviral therapy and associated factors among HIV infected children in Ethiopia: unannounced home-based pill count versus caregivers’ report

Silenat Birosaw1, Woldaregay Erku Abegaz2, Markos Abebe3, Workeabeba Abebe4, Mulugeta Belay2*

1Silenat Birosaw, Tikur Anbessa Hospital, Addis Ababa University, P.O.Box 1176, email: silenatbw@yahoo.com
2Woldaregay Erku Abegaz, Aklilu Lemma Institute of Pathobiology, Addis Ababa University, P.O.Box 1176, email: woldearegay.erku@aau.edu.et
3Markos Abebe, Armauer Hansen Research Institute, P.O.Box 1005, email: markosabebe@yahoo.com
4Workeabeba Abebe, Department of Pediatrics, School of Medicine, Addis Ababa University, P.O.Box 1176, email: workeabebasol@yahoo.com
2Mulugeta Belay, Aklilu Lemma Institute of Pathobiology, Addis Ababa University, P.O.Box 1176, email: mulg2002@yahoo.com

*Corresponding author: mulg2002@yahoo.com
Dear Editor,

We thank the associate editor for reviewing the manuscript. We have addressed the comments highlighted with yellow. Besides, differences in HIV disclosure in table 1 and 3 have been corrected.

Kind regards,

Mulugeta Belay

Corresponding author